Math 132 Winter 2004 R. Schultz

UPDATED GENERAL INFORMATION — JANUARY 9, 2004 The two midterm examinations will be held on Monday, February 2, 2004, and Friday, March 5, 2004.

Here is the first homework assignment, which is due in class on Friday, January 16, 2004. “F&B” refers to the course text by Fraleigh and Beauregard.

Additional Exercise 1 for Section I.A Exercises 2, 10, 22, 36 and 46b beginning on p. 300 of F&B. Additional Exercise 1 for Section I.1 Exercises 2, 6, 22 and 26 beginning on p. 315 of F&B. Additional Exercise 1 for Section I.2